Author's response to reviews

Title: Measuring team factors thought to influence the success of Quality Improvement in primary care: a systematic review of instruments

Authors:

Sue E Brennan (sue.brennan@monash.edu)
Marije Bosch (marije.bosch@monash.edu)
Heather Buchan (Heather.Buchan@health.gov.au)
Sally E Green (sally.green@monash.edu)

Version: 2 Date: 10 July 2012

Author's response to reviews: see over
13 June 2012
Editors-in-Chief, Professor Martin Eccles, Professor Robbie Foy and Dr Brian Mittman
Implementation Science
BioMed Central

Dear Dr Wilson

On behalf of my co-authors, Marije Bosch, Heather Buchan and Sally Green, I am pleased to submit a shortened version of the manuscript “Measuring team factors thought to influence the success of Quality Improvement in primary care: a systematic review of instruments” for your consideration in the journal Implementation Science.

We have shortened all sections of the manuscript, with the most substantial revisions to the background, methods and results (reducing the overall length by about 1700 words). In the background, we have cross referenced to the companion measurement review paper (under review at Implementation Science, MS ID: 8828426636528229) for more detailed description of the conceptual framework used in both papers.

We look forward to hearing from you.

Yours sincerely,

Sue Brennan (on behalf of Marije Bosch, Heather Buchan and Sally Green)